Author’s response to reviews

Title: Management of first metatarsophalangeal joint osteoarthritis by physical therapists and podiatrists in Australia and the United Kingdom: a cross-sectional survey of current clinical practice

Authors:
Kade Paterson (kade.paterson@unimelb.edu.au)
Rana Hinman (ranash@unimelb.edu.au)
Hylton Menz (h.menz@latrobe.edu.au)
Kim Bennell (k.bennell@unimelb.edu.au)

Version: 2 Date: 03 Mar 2020

Author’s response to reviews:

Dear Editor,

Thank you for the opportunity to revise and resubmit our manuscript. We have responded to Reviewer #1’s comments below, and submitted a revised manuscript with changes tracked. We hope that these will be satisfactory.

Thank you once again for the opportunity to have our work published in the Journal of Foot and Ankle Research.

Best regards,
Kade

Reviewer #1
Thank you for your response to the reviewer feedback, the subsequent changes have improved the article. I have noticed 2 inconsistencies regarding the use of abbreviation of Osteoarthritis in lines 148 and 268

RESPONSE: Thank you, we have abbreviated these as advised.

Thank you for adding the sentence referring to the heel raise test and Jack test please add references to substantiate your claim (line 271 to 274).
RESPONSE: We have now added the following references to support the claim that there is “limited evidence” for the use of Jack’s test and the heel raise test respectively:
